Author's response to reviews

Title: Trochanteric osteotomy versus posterolateral approach Function the first year post surgery

Authors:

Margot van der Grinten (m.vandergrinten@erasusmc.nl)
Max Reijman (m.reijman@erasusmc.nl)
Frans C. van Biezen (f.vanbiezen@erasusmc.nl)
Jan A.N. Verhaar (j.verhaar@erasusmc.nl)

Version: 5 Date: 5 May 2011

Author's response to reviews: see over
Dear Editor,

First of all, we would like to thank both reviewers for the comments to improve our paper.

Based on the results of our study we calculated the number needed to perform a non-inferiority study. Per clinical outcome (HHS and 3 subscales of the WOMAC) the numbers were calculated with an alpha of 0.05 and a power of 0.8 (1 – β). This was included in the revised manuscript (abstract page 2, discussion page 11, conclusion page 13).

Yours sincerely (on behalf of all co-authors),

Margot van der Grinten